Author's response to reviews

Title: Intestinal ischemia following laparoscopic surgery

Authors:

Waleed Al-Khyatt (mzxwa@nottingham.ac.uk)
James thomas (driithomas@gmail.com)
David J Humes (david.humes@nottingham.ac.uk)
Dileep N Lobo (dileep.lobo@nottingham.ac.uk)

Version: 2 Date: 8 July 2012

Author's response to reviews:

A case series has been added to the title (highlighted).
Cases were described as case 1, case2 and case 3 in both abstract and case presentation.

As far as patients' consent is concerned, we are working towards achieving consent for each living individual (moved to another geographical area) and also from next of kin's consent for the deceased one. We here also confirm that each patient’s identity and information were completely kept anonymous in writing this report.

Regards,

Waleed Al-Khyatt